



Review

State of the Art Robotic Grippers and Applications

Kevin Tai, Abdul-Rahman El-Sayed, Mohammadali Shahriari *, Mohammad Biglarbegian and Shohel Mahmud

School of Engineering, University of Guelph, Guelph, ON N1G 1Y4, Canada; ktai@uoguelph.ca (K.T.); elsayeda@uoguelph.ca (A.-R.E.-S.); mbiglarb@uoguelph.ca (M.B.); smahmud@uoguelph.ca (S.M.)

* Correspondence: mshahria@uoguelph.ca; Tel.: +1-519-824-4120 (ext. 56044)

Academic Editor: Huosheng Hu

Received: 12 March 2016; Accepted: 31 May 2016; Published: 17 June 2016

Abstract: In this paper, we present a recent survey on robotic grippers. In many cases, modern grippers outperform their older counterparts which are now stronger, more repeatable, and faster. Technological advancements have also attributed to the development of gripping various objects. This includes soft fabrics, microelectromechanical systems, and synthetic sheets. In addition, newer materials are being used to improve functionality of grippers, which include piezoelectric, shape memory alloys, smart fluids, carbon fiber, and many more. This paper covers the very first robotic gripper to the newest developments in grasping methods. Unlike other survey papers, we focus on the applications of robotic grippers in industrial, medical, for fragile objects and soft fabrics grippers. We report on new advancements on grasping mechanisms and discuss their behavior for different purposes. Finally, we present the future trends of grippers in terms of flexibility and performance and their vital applications in emerging areas of robotic surgery, industrial assembly, space exploration, and micromanipulation. These advancements will provide a future outlook on the new trends in robotic grippers.

Keywords: robotic grippers; grasping; end effector; micromanipulators; robots

1. Introduction

The robotics industry was originally developed to supplement or replace humans by doing dull, dirty, or dangerous work. They are used in applications such as: automated assembly lines, observing radiation zones, minimally evasive surgery, and space exploration. Modern industrial robotic arms excel over humans in many tasks. They are capable of lifting 1000 kg [1], are repeatable to 10 μ m [2], and are faster with accelerations up to 15 g [3]. Additionally, the cost of robotic grippers is decreasing while manual labor costs are increasing. This has encouraged industry and academia to develop more advanced robotic arms and grippers addressing both form closure and force closure [1], two of the main components of any robotic gripper. Robotic grippers, by directly being in contact with the workpiece, are tasked with interaction with the environment and grasping objects like human hands for human manipulation [2].

Advanced grasping of complicated objects is an active research area, which includes soft fabrics, microelectromechanical systems (MEMS), and synthetic sheets. In addition, grippers are being designed with different materials. This includes piezoelectric crystals, shape memory alloys (SMAs), magnetorheological (MR) fluid, carbon fiber, and many more. Recent research has also considered bio-inspired gripping mechanisms, which leverages nature to develop products that solve problems that are more industrial.

Previous papers have reviewed robotic grippers with more emphasis on specialized applications. In [3,4], the use of grippers was covered for automated production processes. Different gripping systems, such as artificial vacuum, magnetic and mechanical gripping, were discussed in [5].

Robotics **2016**, 5, 11 2 of 20

In [6], only parallel manipulators gripper mechanisms were covered. In [7], grippers for surgical applications can be found. In [8], only dual arm manipulation was covered. In [9], a review of space robots was discussed. In [10], plant production robots were reviewed. In [11], only the contact strategies for micro components were examined. In [12], end effector control schemes were discussed. Some modern examples of grippers include the Kuka KR 1000 Titan; Dexter on the International Space Station; the Explosive Ordnance Disposal manipulator on the iRobot 510; and the da Vinci Surgical manipulator [13–16].

There are numerous applications of where robotic grippers can be used. This highlights the importance of reviewing the many recent developments that are occurring in research and discuss the future directions of designs and applications. This paper provides a broad perspective on the different types of grippers; our paper is especially focused on applications of grippers that have not been covered in previous survey papers including grippers for deformable and fragile objects and living organs. The earliest types of grippers in our survey are discussed along with the improvements made to them, and most importantly, the recent advancements. We discuss different categories of grippers being used for known and unknown environments, medical and industry applications, and micro and nano manipulation. We also compare new grippers' designs and state their advantages and limitations. Our paper focuses on emerging applications such as in medical, agriculture, micro and nano manipulation, new design developments in this field, and future direction of design, technologies and applications of the trending grippers' designs, which differs our work from other survey papers. We believe this paper helps researchers and industry discover newer grasping mechanisms, their advancements as well as challenges associated with them. We conclude our work by discussing the challenges and future directions in this field.

The rest of this paper is organized as follows: Section 2 covers the developing grippers organized by application, Section 3 presents different gripper designs, Section 4 discusses the future direction and discussion of robotic grippers, and Section 5 concludes the paper.

2. Applications of Robotic Grippers

2.1. Grippers for Industry

The earliest grippers were first developed for industrial applications. They are commonly defined as grippers used for mass production purposes that are mounted on a stationary platform. The industrial grippers can be studied through different aspects such as geometrical condition of grasping, position and orientation of grasping, static equilibrium of grasped object, and dynamic conditions [17,18]. We mainly focus on the performance, adaptability and flexibility of the grippers.

The first industrial robot was the UNIMATE installed in a General Motors assembly plant in 1961 [19]. This was a rigid parallel manipulator that grasped hot pieces of die cast metal. Since then, many companies have embraced robotic gripping technology and have developed different drive mechanisms [20–22]. These were typically driven by electric motors or hydraulic actuators, but more recently piezoelectric and shape memory alloys are being used for actuation.

Industrial grippers can be split into different categories such as grippers used in known environments and grippers used in unknown environments.

2.1.1. Grippers for Known Environments

Grippers that are used in known environments typically have parts that come on an assembly line. The parts are positioned in predefined orientations, which make it easier for the gripper to pick up the object. These grippers can use servos, non-contact, contact, or a combination of sensors for feedback. Sensors can include hall sensors, accelerometers, ultrasonic, or photoelectric sensors to name a few. This can be used for detecting many variables such as: position, force, torque, velocity, and acceleration [23]. These sensors prove useful in many situations such as detecting whether objects are being grasped, or by providing information to a supervisory computer to control an assembly line

Robotics **2016**, 5, 11 3 of 20

process. Two examples of commercialized grippers are the Schunk DPG-plus and the Robotiq 2-Finger Gripper [24,25].

In academia, grippers are prototyped with many different types of sensors and feedback mechanisms. In [26,27], three sensors are used to detect deflection in each axis (x, y, and z). In [28], the sensors are built from materials such as conductive silicone. In [29], sensor feedback is integrated into a smart sensor network where each node performs signal acquisition. New developments have also encouraged the use of a capacitive force sensing array for the artificial skin of a robot [30]. Tactile sensing systems have also been incorporated into prosthetic hands [31]. It uses a biomimetic tactile sensor, which is capable of triaxial force sensing. In [32], a novel tactile sensor is presented using flexible piezoresistive rubber.

Proximity sensors have also been mounted on grippers that use an active prediction, planning, and execution system for intercepting moving objects [33]. It moves to the earliest pregrasping location close to the object and the proximity sensors utilize a fine-motion tracking algorithm to pick the object. This removes the need for continuously tracking the motion of the objects during the planned movements and it reduces the distance between the object and end effector.

A shell shaped end-effector is designed in [34] where the force can be sensed on any position of the body. The position and force of the contact can also be measured in six axes.

There are also claw like end-effectors that can be thrown using a launch platform. In [35] the claw is launched using a swinging motion while the range is measured by a laser range sensor. One end of the claw is attached to the base by a string. The tension of the string is adjusted midflight, which allows the claw to fly in parabolic trajectories to the target.

In [36], a minimally invasive surgical gripper was developed. The proposed design is advantageous as it enhances the surgeons' dexterity. The prototype gripper is a parallel end effector controlled by a joystick. The position and orientation is acquired from a motion tracking system by Northern Digital Inc. and physical movements are displayed using a screen in real time.

Control systems are also being studied which can significantly improve the tracking and performance of grippers. In [37], a 3-DOF parallel manipulator was designed with PID and fractional-order PID to improve the tracking performance. Servo motors were used and it was found that the fractional-order controller reduced the transient and steady state error. In [38], an oscillatory-base manipulator was tracked using H_{∞} and PD controller. It used servo motors and it was found to be superior to a PID controller.

Underwater grippers with feedback systems have also been developed [39]. For this design, a multi-limb gripper with tactile sensors, which is capable of sensing objects up to 6000 m in depth. Testing the gripper system in a pressure chamber at 600 bar demonstrated the effectiveness of the deep sea system.

In [40], six position sensors and six servo valves were used as feedback for a 6-DOF electrohydraulic parallel robot. It utilized decoupled-space control (DSC) to further improve the performance of the robotic gripper. It was found that the DSC framework improved the performance by solving the dynamic coupling effects of the multi-DOF parallel robot.

Grippers also include attractive grasping, also known as astrictive grippers. This attraction can be: vacuum suction, magneto-adhesion, or electro-adhesion. Some of the earliest astrictive grippers were used for sheet metal fabrication [41]. A theoretical model of a suction gripper is found in [42]. This model is then expanded to multi-link manipulators in [43]. They are also used heavily in the packing industry as astrictive grippers have a soft grasp on both heavy and light objects [44]. This is used in a modular vacuum conveyor system to pack objects neatly in trays after they come off the assembly line [45]. They are advantageous as they can pick up slippery, oily metal sheets [46]. Another common application is picking up large glass panes during the manufacturing process [47].

In [48], three robotic packing cells were utilized to pack soft drinks into boxes. The end-effector could pick up 12 soft drinks at a time with a vacuum suction cup for every drink. Once the packing robot was installed, the assembly line was able to operate 25% faster. The workers were then redeployed to different areas of the company to add more value to the end product.

Robotics **2016**, 5, 11 4 of 20

In [49], a flexible assembly station is developed for aerospace applications. A great difficulty in manufacturing fuselages and wings for airplanes are the changing surfaces. This designed assembly station consists of only 8 actuators with suction cup grippers. These are fitted with load cells and the actuators are extended slowly to search for the fuselage. Once it finds the part, the suction cups are activated and the suction cup heads are locked in position. The advantage of this holding mechanism is that every actuator can map the forces. It can also provide valuable information during manufacturing processes as excessive loads will allow the process to be stopped immediately.

Some of the challenges that suction cups have are the speeds that the work pieces can be moved at. If the acceleration of the work piece is too fast, the gripper could lose suction. To increase production speeds, more innovative methods of holding these objects must be developed. There can also be challenges associated with the speed of deceleration such as during emergency stops. A proposed solution to this is to use friction cups. They have higher coefficients of friction, which can handle large parallel friction forces; however, they have weak friction forces in the perpendicular direction. Vacuum pressure in suction cups is also an issue as centralized vacuum systems add weight, material and costs. It also could affect the suction pressure at different cups since longer tube lengths will cause performance penalties. In all cases, only the small volume between the cup and the vacuum pump need to be evacuated. Miniaturizing such systems can avoid these issues, however in many cases the amount of vacuum pressure created is not large enough to hold the [50].

2.1.2. Grippers for Unknown Environments

In many cases, grippers might be tasked for pick-and-place operations without knowing the conditions of the environment. Different design and techniques have been developed to increase the flexibility of grippers in unknown environment such as using vision systems, sensory feedbacks, and novel mechanism with flexibility in gripping.

Camera systems are used to detect the presence of objects. One of the first parallel grippers designed with camera systems was in [51]. This was built to pick up randomly oriented work pieces from a bin. Since then, increased computing power and finer resolution vision systems were developed for gripping [52]. In order to pick up the object, the operator simply had to point using their finger within the workspace of the robot.

Vision systems were further improved in the semiconductor industry [52]. Large polysilicon nuggets require very careful handling before they are fused into larger quartz crucibles. These nuggets range from 600 grams to a few grams, which need to be carefully packed before heating. Since they are packed using complex rules, an automated system is used. There is an overhead charge-coupled device (CCD) camera and two scanners to map environment.

The gripper of autonomous mobile manipulators is a problem that is being challenged using a modified image-based visual servo controller with a hybrid camera [53]. An eye-in-hand web camera was used to visually track the object in the unstructured environment while a stereo vision camera was used to measure depth information online. The controller combined the information and used a rule base to tune the gain in order to improve the response speed of the controller. This method was then validated using experiments.

In [54], a vacuum gripper was developed with machine vision. Using segmentation algorithms, the best hold position and orientation was determined. The data is then relayed to the robotic vacuum to pick up the object and move it to the desired location.

In [55], a new image-based visual servo controller was created for a 6-DOF gripper. The controller was able to achieve a more linear trajectory and reduced the risk of having image features leave the field of view.

Human grasping has also been investigated where a 3D camera and inertial measurement unit was used to track the movements of a human hand [56]. It incorporated a Kalman filter and an adaptive multispace transformation on the movements of the hand. The gripper would then move in tandem with the hand.

Robotics **2016**, 5, 11 5 of 20

Mobile robot grippers have also been studied using 3D scanning of unknown environments [57]. The gripper was used for autonomous climbing maneuvers. Feedback was also provided by an inertial measurement unit, joint absolute encoders, and pressure sensors.

In [58], grasping maps were developed using camera systems. The map in this configuration utilized online and parallel computations to reduce the time by more than 90% compared offline systems. The method was then validated in a physical environment with both simple and complex objects.

In [59], grasping planning was conducted using a Kinect RGBD camera system, a two fingered gripper, and an industrial robot arm. It required a learned mapping between geometric states and logical predicates. The task was to find the cups on the table, stack them, and put them on a tray. The approach in this case was not entirely complete as some configurations were not able to generate a plan. If there were issues with generating a plan, a proposed solution would be to remove one of the cups off of the table and stack the rest. This work is being continued to execute plan generation on different types of objects.

In summary, the main objective of robotic grippers in unknown environment is flexibility. Different approaches have been addressed to overcome this challenge. It is discussed that 3D scanning and vision feedback are shown to perform well. The other objective is novel mechanism design, which provides flexibility to the gripper inherently. These designs can partly take care of controller task by intelligent structures mechanically.

2.2. Grippers for Fragile Objects

With the improvement of end-effector sensors, the idea of picking up fragile objects was explored. In [60], an end-effector was designed for harvesting lettuce. This design included a machine vision device, six photoelectric sensors and a fuzzy logic controller. The designed end effector was able to harvest lettuce at a rate of 5 s per lettuce with a success rate of 94.12%.

In [61], an enclosed hygienic food gripper was designed with force feedback sensors. One finger on the gripper is stationary while the other finger moves by magnetic attraction. The actuator is placed inside with an inner magnet while an outer magnet moves the finger on the outside of the container.

Another design was investigated which grasped fruits [62]. This end-effector was built to incorporate both clamping and cutting to avoid possible damage to the fruit. A similar design is used to harvest strawberries, which uses a machine vision unit [63].

In [64], a citrus harvester was also designed with a large field-of-view camera. Similar designs were presented in [65,66].

A Bernoulli principle type end-effector was designed for handling sliced fruit and vegetables [67]. It allows the objects to be lifted using the airflow over the surface to minimize contact. This then reduces the probability of cross-contamination and damage to the produce. Another benefit of using this type of gripper is to reduce the amount of surface moisture on the object. This gripper was built and was found to be feasible in handling food products.

A robotic gripper was used in [68] to assess the firmness of mangos, accelerometers embedded in the gripper fingers were used to test the ripeness of mangos. Results were compared to the ripeness index of mangos, and the robotic gripper proved effectiveness to do this task.

A hybrid tomato picking end-effector was also developed using a combined impactive and astrictive design [69]. This was capable of removing tomatoes with short peduncles, however the gripper had difficulties when there were leaves and stems in the way. In [70], a similar gripper was developed which consists of four fingers padded with foam to reduce damaging the fruit. Once the fingers close, it brings the tomato to the center where there is a suction cup, which assists the grip. It was capable of picking tomatoes at an average rate of 74.6 s per fruit with a 95.35% attachment success.

A similar hybrid cherry harvesting robot is designed in [71]. It utilized a 4 DOF gripper, a 3D vision sensor, and a two fingered grasp with a vacuum tube in the center. The purpose of utilizing a hybrid design is that cherries are harvested with the penuncle attached. The end-effector is positioned

Robotics **2016**, 5, 11 6 of 20

around the cherry and the base is vacuumed to the center. The two fingers then grasp the penuncle and both the cherry and the penuncle detach simultaneously.

Deformable objects have also been studied where the robotic gripper evaluates the robot trajectories when colliding with objects [72]. It weighs the object deformation and compares it to the travel costs. The deformation models can be used to perform FEA, and a similar lower computationally intensive model is presented using Gaussian process regression. The planning method is then tested using a wheeled robot and a robot arm.

As different approaches were explored in robotic grippers for fragile objects, it is discussed that the force feedback is a key element as well as flexible designs such as soft grippers not to harm the objects.

2.3. Grippers for Medical Applications

In use of robotic grippers in surgery, one of the main issues is the lack of force feedback and damaging the biological tissues. Soft bodied grippers are very suitable in the medical field based on their self-limiting and intrinsic safety features, which provides safe interaction with biological tissues.

In [73], a soft gripper design is developed for delicate and safe interaction in minimally invasive surgery. As reported, this design could enforce maximum force of 1 N. They used elastomeric material and discussed that their design is easily scalable.

A robotic gripper with viscoelastic force fields control was developed in [74]. A two finger with precision grip control design was designed and analyzed in this work and they showed that thumb and finger force are highly correlated. Precision grip control provides adaptation capabilities which helps to grip objects based on their mechanical properties. Although this design gave insight on the ability to adapt to unknown precision grip dynamics, the design is limited to linear 1-D gripping and no result was reported for performance of gripping of nonrigid objects.

Robot-assisted surgery is another application of robotic grippers. In robotic surgery safety and autonomous control is essential because of injurious result of uncertainties and disturbances such as time delay in tele-surgery.

In [75] a star shaped micro-gripper was designed and developed with the application of tissue excision. By performing experiments within a live animal they showed that these developed micro-grippers are capable of excising tissue samples from real organs and hard to reach places inside a body. These grippers are fabricated using conventional multilayer microfabrication and actuated with magnetic field.

A design of a soft robotic gripper for manipulation in minimally invasive surgery was presented in [73]. In this design, the authors used only soft materials and an under actuated system to adapt the finger shape and apply a specific amount of force. This gripper was considered a surgical tool and it was made of soft material for safe interactions. In addition, such soft gripper designs are safe for surgery due to limited soft gripper force.

Suction techniques can also be used in medical applications [76]. This gripper was designed to grasp the bowel, which is a large, delicate, flexible, and slippery part of the body. This suction method was found to be capable of firmly gripping bowel sections: however, it is not known whether a manual pump or vacuum pump would be more optimal.

A miniature robotic gripper was designed in [77] for retraction tasks in minimally invasive surgery. They showed that the gripper can generate a maximum gripping force of 5.3 N. The design of this gripper allows performing within a narrow access port. They integrated brushless motors to enable additional degrees of freedom by magnetic anchoring and improving dexterity of overall platform.

In [78] a 4 degrees of freedom surgical forceps integrated with force sensor is developed for minimally invasive surgery. The pulling and grasping forces can be measured for haptic feedback control. Their design was experimentally validated for the open-source surgical robot platform Raven-II.

Robotics **2016**, 5, 11 7 of 20

In [79] a soft pneumatic chamber gripper device was designed for tissue manipulation. This gripper can pick up objects up to 2 mm with a gripping force of three times smaller than the grip force of forceps grippers, which prevents tissue damage during surgical manipulation.

In [80] a magnetically guided and actuated milli-gripper was developed for medical applications. It was shown than using permanent magnets they were able to control the grippers in both tethered and untethered configurations. This design is an integrated capsule of an electromagnetic coil with a soft magnetic cobalt iron core and a magnet.

Recent advancement in medical grippers made them more reliable to be used in applications such as robotic surgery, minimally invasive surgery and medical grippers. The recent improvements are designing novel mechanisms, developing and employing high tech actuators. Although many works have addressed force control problem in medical application, the challenge of force control still persist.

2.4. Micro and Nano Grippers

Different designs and technologies have been developed for micro and nano gripping by employing advancements in micro and nano electromechanical devices.

MEMS were developed to grip micro-sized objects. These have been used in the semi-conductor industry where parts are assembled on a wafer substrate [81]. They are also used for manipulating bio-and nano-materials [82].

A hybrid MEMS microgripper was designed in [83]. It utilized an electrostatic comb drive microgripper with an integrated vacuum tool. The tip of the arm was able to deflect 25 μ m and it was able to pick up objects ranging from 100 μ m to 200 μ m.

A thermal bimetallic actuation gripper was also developed in [84]. The design of this gripper was unique as it combined nanofabrication techniques with conventional microlithography methods to create a gripper that was 100 nm and less in size.

A microgripper is also designed using an electrostatic linear motor [85]. This motor consists of a scratch drive actuator, which was able to move in micropositions with high accuracy. It was also modelled using FEA and the fabrication process is covered. The electrostatic actuation is successfully obtained with high precision and low electrical consumption, which is capable of being integrated with integrated circuits (ICs).

Another technique is utilized to microfabricate grippers to sub-micrometer dimensions [86]. The process utilized a thin SiO_2 layer of a silicon-on-insulator wafer to create the gripping tips. It was able to grip 100 nm gold spheres.

In [87], a novel electrostatic MEMS gripper with integrated capacitive contact sensor was developed. It is able to grip objects as low as 12 μ m at a driving voltage of 55 V. While this is occurring, the transverse comb differential capacitances act as a contact sensor to prevent damage. It also utilizes vibrations to release the parts.

In [88], a polymer made from SU-8 was used to build microgrippers. It included tensile force sensors and were actuated by using an electro-thermal effect. Large displacements of the gripper were possible up to a maximum of 100 μ m for up to 50 cycles. Variable forces are also achieved from a few μ N to 1 mN.

In [89], a two fingered microhand is developed for high speed gripping. This microhand was able to pick-and-place objects 40– $60~\mu m$ in size within 1 s.

A combined vacuum gripper, two-camera vision system and 4 DOF robotic gripper was designed in [90]. This design was compared to a standard vacuum gripper (standard dispensing needle) and a multi-lumen nozzle. In these tests, it was found that the grasping of the microcomponents was always successful. Releasing the microcomponents however was poor in some conditions. Further modeling for this gripper will be developed to determine the behavior during the release stage.

In [91], a miniature polycrystalline silicon gripper was built. This ingressive gripper was designed using a snap-and-lock configuration where a male and female portion would click. The male portion is shaped similar to an arrowhead while the female portion is shaped with two inner hooks. When they

Robotics **2016**, 5, 11 8 of 20

snap together, a tensile load is applied to move both sections. The assembly is tested using a scanning confocal Raman imaging which confirms that the local peak tensile stress is approximately 50% of the lower bound material strength. This stress mapping is also covered in [92].

In [93], an electrostatic gripper is developed. The mechanism was able to grasp standard microcomponents and delicate mini and micro parts. It also added new theoretical and FEM models, which were justified in the design of the gripper. It was able to grasp and release different material shapes with various dimensions between 0.3 mm and 1 mm. It was found that the reliability of the grasp is close to 100%.

An integrated circuit transfer robot was developed using a Gecko-inspired microfiber array in [94]. The Gecko-inspired method is investigated as vacuum grippers can damage wafers if the pressure deforms the chip excessively. This gripper utilizes the friction force caused by the mass of the wafer on the end-effector. The four bumps of microfiber material is added in a square configuration which was found to have the least amount of deformation. This gripper was able to operate at an acceleration of 4.155 m/s^2 which is much faster than using rubber or stainless steel bumps.

Contigutive grippers have also been designed to use a liquid water layer between the manipulator and the object [95]. The manipulator freezes the water to produce ice, which adheres to both surfaces. This was also designed for submerged applications [96–98].

As discussed, microgrippers have been of an interest to researchers in recent years because of their vital application in micro fabrication, micro assembly and micro manipulation. Recent advancements in microcomponents and more specifically MEMS as discussed in the literature makes this technology cheaper, more reliable and easier to implement.

2.5. Soft Fabric Grippers

An ongoing challenge in gripper design is in picking up fabrics. In order to do this, penetrating grippers have been designed which are also called ingressive grippers. They are used in the textiles industry to hold fabrics since suction cups cannot vacuum to the material due to the porosity. Textiles however can be penetrated while being moved with minimal underlying damage to the woven structure.

One of the earliest designs used a spiked wheel, which separates different sheets of textiles from each other [99]. This was improved using a different style of gripper in [100]. These types of grippers are also used to remove backing sheets from fibers [101]. Fortunately, most industrial ingressive grippers can adjust these variables to prevent damage to the material [102].

Pre-peg carbon fiber sheets have also been grasped in [103]. Picking up continuous carbon fiber sheets has also been investigated in [104].

A suction cup gripper was additionally developed for fabrics [105]. This is difficult because the fabric is not allowed to be deformed during the automatic or manual cutting process. The end-effector consisted of a flat surface with 0.5 cm holes for suction and outer 0.1 cm holes for positive pressure. This gripper did not have tactile feedback sensors, which could improve part handling and manipulation. The model for this specific gripper is covered in [106].

There have been improvements made to suction cup grippers, which are now able to pick up C and L-shaped profiles. It is able to do this by unrolling the fabric in different stages. In addition, textiles can be shaped to different geometry due to modifications in the end-effector shape [107]. In this particular case, the carbon fiber textile was picked up around a cylinder and it was draped around contours to cure [108]. A highly flexible and moldable suction cup was also designed to pick up leather plies [109]. This flexible cup was advantageous since it did not leave imprints on the leather compared to other commercial vacuum cups.

Other methods of picking up fabrics have also been used such as direct contact. They are called contigutive grippers, which work by using chemical adhesion such as glue and sticky adhesives. This can also be done thermally as well by melting a surface to adhere to the object. For the textile industry, a permatack adhesive is used with a contact pad [110]. The disadvantage to this method is that microfibers will adhere to the surface over time, reducing the ability to stick to textiles.

Robotics **2016**, 5, 11 9 of 20

Thermal adhesion is also used in carbon fiber sheets [111]. This works by thermally melting the resin in the carbon fiber sheet to achieve a solid bond. Improvements since then have lowered bonding time to less than one second [112]. This is compared to chemical curing which can take over ten seconds.

In summary, suction grippers have been shown to perform well in grasping soft fabric objects. However, different methods are being developed to employ this technique such as adhesion.

3. Design of Robotic Grippers

In this section, we study the design peculiarities of grippers with different applications and given tasks' purposes. The grippers' characteristics such as dimension, weight, rigidity, and simplicity are considered in this section for different designs of robotics grippers. The main factors that need to be considered in the study of design peculiarities are the characteristics of the gripper, the characteristics of the objects, gripper technology, flexibility of the gripper, and cost of the designs. In addition, other factors including grasping force, grasp configuration and transmission characteristics of the device are studied.

3.1. Piezoelectric Grippers

As piezoelectric materials are decreased in price, research has focused more on utilizing piezoelectric grippers for active manipulation [113]. Some benefits of piezoelectric grippers are the simplicity, the ease of use, and the low power consumption compared to mechanically actuated grippers.

Some of the earliest grippers were made out of piezoelectric polymeric polyvinylidene fluoride (PVDF) [114]. This was improved in [115] where the design was arranged in a rhombus configuration. This amplifies the force applied to the object and it allows one of the piezoelectric strips to act as a force sensor. Regardless of this, the prototype was only able to apply 1.0 N at the tip. In [116], a piezoelectric tweezer was designed for use in robot assisted surgery that is MRI compatible. It is amplified using a nested strain mechanism, which increases the displacement of the tips to useable levels. The tweezer tip position and force was shown to be within a 12% average error between the self-sensed and true values.

In [117], a voltage of $(\pm 60 \text{ V})$ was applied on a piezoelectric actuator. This causes unstable behavior so a PID controller was designed to supply the voltage to achieve stability. It was found that using a PID voltage controller helps with precise micro manipulations. This work is continued in [118], where a PD controller is used instead. In [119], a PI-based sliding mode controller is developed for a piezoelectric actuator. The proposed controller was able to provide better tracking, faster responses, less chatter and higher positioning precision.

Piezoelectric bimorphs have also been analyzed theoretically and experimentally in [120]. The study focused on the transient response and the frequency response with various input voltages and signals along the length of the actuator. This gripper could manipulate objects down to 50 μ m with the only the actuator itself.

As discussed, although different designs were proposed with different control approaches, the main challenge that still persists is the position control and stability of piezoelectric grippers.

3.2. Multi-Fingered Grippers

Many approaches have been developed to increase the flexibility of the parallel gripper. By using multiple fingers, the available motions for the robot are increased. One of the earliest three-fingered hands was developed in [121]. They have also been integrated with feedback sensors in [122]. This design utilized 87 touch sensors that were distributed over the surface of the end-effector. Some examples multi-fingered grippers include the Robotiq 3-Fingered Hand, the IH2 Azzura Hand and the Shadowrobot Hand [123–125].

Recent end-effectors studied a wide range of possibilities to improve typical parallel grippers. In [126], a fiberless, flexible microactuator (FMA) was developed. The FMA gripper consists of hollow cylindrical chambers along the length of the gripping finger. When the chambers are subjected to a positive or negative pressure, the chamber expands and contracts which bends or extends the fingers. The model that was built was capable of holding fish eggs without breaking.

In [127], a novel design and actuation principle was implemented to explore grasping capabilities. The design consists of a three fingers robotic gripper, the fingers were made of soft materials to increase adaptability. An under-actuated mechanism has been used using a wire loop actuation system. Grasping and holding capabilities have been tested on different sizes and shapes. Due to the adaptability of this design, it is considered good application for morphological computation principles in bio-inspired robots designs.

A Chinese massage end-effector was designed in [128]. This was able to thumb knead, press, roll, vibrate, and pinch. Visual tracking was used as the patent wears a massage vest with highlighted pressure points. When the end-effector is working, the patent will set a pain threshold, which tells the robot to not to exceed the limit by using force-position control.

In [129], a DARPA ARM robot was enhanced with a framework to allow the end effector to control position and force with either visual of tactile feedback. The three fingered grasp, was capable of grasping a free tool such as a drill while accounting for the uncertainty in the hand-tool interface.

Multi-fingered graspers have also been used to fry pancakes [130]. Two robots called TUM James and TUM Rosie were sent instructions over the internet. They then generate robot action plans, which involve operations such as opening drawers, retrieving pancake mix and transferring items. These tasks are done perceptually using a semantic 3D object map of the environment. They draw from a database of 3D models, which locates and identifies the objects quickly. Multi-fingered grippers have many applications such as in grapping oddly shaped and unknown objects. Several actuation systems are presented such as flexible microactuator, wire loop actuation system; however, it is suggested that soft materials may provide more adaptability to these grippers.

3.3. Enveloping and under Actuated Grippers

Grasping oddly shaped objects has been an ongoing challenge in end effector design. Some grippers have focused on enveloping and underactuated mechanisms. This was the case in [131] where a pair of underactuated hands was built. This type of gripper is valuable due to the adaptability of the gripper to mold around the object it is holding. A similar paper in [132] also developed underactuated grippers for a similar application.

An enveloping gripper has also been modeled after the biting movement of a snake mouth [133]. The snake mouth design is capable of expanding and opening up to an angle of 150° . The advantages of this design allowed continuous clamping of smooth edged objects. It was also able to grasp items bigger than the end-effector itself.

In [134], a bio-inspired method of contacting objects is modeled after a chameleon shooting its tongue. A permanent magnet end-effector is attached to a string, which is loaded into a catapult. During testing, the catapult end-effector was successful 92% of the time when capturing a falling object 0.7 m away. This end-effector has advantages as it can reach objects, which are not in a clear line of sight from the base. By firing the end-effector above or around the obstacle while constraining total length, the magnet can drop behind or loop round the obstacle to attach to the target.

Similar to the underactuated hands, a set of parallel, fingertip and enveloping grasps was designed in [135]. It is controlled by a single actuator, which pulls a tension cable along the fingers. This design is capable of grasping objects in multiple configurations. The links will flex based on whether the object contacts the distal or proximal links.

In [136], a pneumatic actuator was developed. It consists of an outer skin which is pumped full of air. Since one side of the skin is thicker than the other, the weaker side expands and bends the thicker side inwards. This pneumatic actuator, when subjected to moderate pressures (<1.25 atms) could bend

upwards of 180° . Other methods to grasp oddly shaped objects, which are used for deforming surfaces are malleable grippers. One of the first deforming elastic layer grippers was built in [137]. This is further investigated in [138] where a pair 3-DOF fingers is modeled with soft tips.

Enveloping grippers are mainly biologically inspired grippers and recent advancements showed that by developing such novel mechanisms we will be able to implement grippers more autonomous with less control effort compared to multi-fingered grippers.

3.4. Malleable Grippers

Malleable grippers are made from materials that change viscosity. They are typically made from using a flexible outer skin with materials inside being either: electroheological (ER) fluid, magnetorheological (MR) fluid, or pellets. The outer surface is pushed up against the object and it molds to the geometry. The inside is then hardened to provide an impactive force to hold the object. When the object needs to be released, the inside will return to a more fluid like state to allow for free movement.

These soft grippers are mainly used in shape-adaptable grippers, which are not only soft, but they are strong and versatile as well. Several soft grippers' designs have been addressed in the literature with variety of technologies. Soft grippers can be adapted to different shapes where rigid grippers are limited.

One of the earliest malleable grippers used ER fluid [139]. The ER fluid changes its viscosity in response to an electric field. When an electric field is present, the ER fluid behaves like a Bingham plastic. When the field is removed, it behaves like a liquid. This gripper was tested and was capable of handling fragile objects such as eggs. The disadvantages are that using ER fluid requires a lot of energy and the gripper is susceptible to electric arcing.

A much safer malleable gripper was designed using granular material [140]. To change the viscosity of the material, a vacuum is applied and the gripper remains rigid. In [141], the testing and use of the skin gripper is performed. The reliability is increased to 85% and the error tolerance is increased to 25%. An added feature was that this vacuum could be reversed, and capability of shooting objects out of the mold is demonstrated. A commercialized version of this design is now called the Versaball [142].

MR fluid grippers have also been designed to pick up fruits in [143]. The MR fluid changes its viscosity in the presence of a magnetic field. The MR fluid is placed in two pouches near the end of the arms, which contacts the fruit. When the magnetic field is activated, the MR fluid changes in viscosity and the pouches mold around the fruit. With this design, the fruit can be lifted with low force without damage. An advantage to this design is that the amount of pressure applied on the outside of the fruit can vary based upon the magnetic field. This design was satisfactory, however the MR fluid is very expensive.

In summary, malleable grippers are highly flexible in grasping objects with different shapes; however, the dexterity of these grippers is still a challenge of design.

Table 1 compares the characteristics and performance of grippers with different designs that were discussed in this section.

As discussed in literature, we can compare the ability of the gripers to grasp different types of objects based on the categories of Impactive, Ingressive, Astrictive and Contigutive. As presented in Table 2, impactive grippers have been used in applications of picking up different types of objects from solid flat to oddly shaped and fragile. Astrictive grippers, similar to Impactive grippers, have been used for different types of objects more commonly for fragile and irregular objects. No results have been reported, to the best of our knowledge, regarding employing Astrictive grippers for irregular shaped objects. Ingressive grippers have been used in grasping flexible sheets and in some in MEMS assemblies. Contigutive grippers have been used in grasping solid flat objects, rigid and flexible sheets, and MEMS assemblies.

Design	Advantages	Drawbacks	Significant Application	Example of Used Material and Actuators
Piezo-electric Grippers [113–120]	Simplicity, Ease of use, Gripping small objects (down to 50 μm)	Low gripping accuracy	Micro and nano gripping	polymeric polyvinylidene fluoride (PVDF)
Multi-Fingered Grippers [121–129]	Flexible gripping for different object shapes, Gripping with force feedback	Control complexity	Grapping all shaped objects with force control	Soft materials, flexible micro actuators (FMA), wire loop actuation systems
Enveloping Grippers [130–137]	Adaptability to mold around the object	Low force control capability	Grapping oddly shaped and unknown objects	Pneumatic actuators, cable-driven under-actuated mechanisms
Malleable Grippers [138–142]	Adaptable to different shapes, reliable gripping	Low gripping dexterity	Grasping unknown and specially deforming objects	MR fluid, ER fluid, granular material

Table 1. Comparison of the ability of grippers to pick up different types of objects.

Table 2. Comparison of the ability of grippers to pick up different types of objects.

	Type of Grippers				
Types of Objects	Impactive	Ingressive	Astrictive	Contigutive	
Solid Flat Objects					
Solid Curved Objects					
Solid Irregular Shapes					
Flexible Sheets					
Rigid Sheets					
Fragile Objects					
MEMS Assemblies					

4. Future Directions and Discussion

Recent developments in terms of actuators and sensors technologies as well as material science have made the grippers more reliable, faster, safer, and more robust. These advances have resulted in introducing new applications such as new locomotion for mobile climbing robots (e.g., JPL's Rock Climbing Robot), hopping robots, space satellites, underwater robots being used in exploration and pipeline repair, high speed manufacturing, and robotic surgery. These development have opened doors for new research on employing new materials and designs as well as incorporating new technologies some of which are summarized as follows:

- Adaptive and self-adaptive grippers: these grippers have a great potential to provide flexibility in grasping objects with different shapes in industrial systems such as Festo PowerGripper, Finger Adaptive Robotiq, SARAH in international space station.
- Modular grippers: they use standard components such as finger type grippers, vacuum cups, and
 locating pins to construct complex grippers. These grippers have been used in applications where
 high performance and flexibility are required such as assembly in space. They can accommodate
 change in physical, geometrical, chemical, mechanical properties of the objects significantly by
 employing different standard gripping components.
- Reconfigurable grippers: these grippers have the ability to change into different specified
 configurations and pick different objects. These grippers have applications in automotive industry
 and space robotics.
- Smart material based grippers: These grippers use smart materials for grasping objects with different shapes such as grasping by particle jamming (e.g., granule-filled bag), electrorheological (ER) fluids, Giant ER Fluid, ER fluid with electroadhesion, pneumatic actuators, and shape memory foams. Although these grippers have been used in industry for a long time, due to their simple

actuation mechanism and low weight, employing this technology for robotic grasping is still challenging because they have lower gripping forces compared to conventional grippers, they are mostly slow actuators, and there is a control problem in precision actuation of these materials. There are ongoing research to increase the gripping force and precision accuracy. Some of these include developing controllers such as repulsive force control, sliding mode control, and ANFIS controllers. It is worth mentioning that electrostatic attraction provides more dexterity since they use film like layers.

- Novel mechanism design grippers: These designs provide inherently flexibility with a minimum required supervision by incorporating smart mechanisms such as bionic handling assistant into the grippers. The main objective of these designs is to have high performance with less control effort.
- Soft grippers: Different designs of soft grippers have been developed such as electroadhesion grippers, single and multi-segment grippers, artificial muscle soft robotic grippers have been developed. These grippers have been able to mimic human's hand. Flexible, microscopic hand-like gripper can help surgeons to remotely guide surgical procedures or perform biopsies. Most of these designs utilize soft robotics and artificial skins for simpler control and passive adaptation. Soft materials enable gripping automation beyond the capacities of current technology. One of the advantages of soft robotic grippers is partially taking care of the control part by the physical properties of soft grippers unlike rigid grippers. However, introducing softness into the design of grippers requires new set of design and control principles compared to hard grippers.

There are ongoing attempts to improve grippers in two-fold: performance and flexibility. Performance indicates accuracy, speed, readability, gripping strength, robustness, and flexibility denotes variety of objects that can be grasped. Most of the challenge in this aspect is whether objects are known/unknown. When one is dealing with unknown objects, the focus is to employ flexible grippers, while in grasping known objects, the focus is on the increasing the performance. Achieving flexibility and performance simultaneously still remains challenging because increasing performance usually results in decreasing flexibility.

To increase flexibility, new develops are made: grippers with human-like fingers are one of the examples, which use control systems with feedback from the interaction between the gripper and the environment to mimic humans grasping. However mimicking human hand requires significant amount of computation and it complex in terms of control. Although development of robotic grippers to mimic human hands is still a challenge regarding sensing and actuation, the use of artificial hands in industrial and social robots will be more common in the coming years. An alternative approach is to use different designs. Some of these designs include malleable grippers, MIT gripper, which is based on jamming of granular materials, and RobotinoXT.

Performance improvement is application dependent. Increasing performance can be achieved through design of robust controllers, implementation of sensors and powerful actuators as well as physical design. In terms of sensing, researchers will continue to mimic the human gripping capabilities. Vision sensing is anticipated to evolve to increase the learning capabilities of robotic grippers with incomplete knowledge of the surrounding environment. In addition, incorporating visual feedback in the design of gripers enables them to effectively communicate with unknown environments making them robust. For actuation, artificial muscles implemented in robotic hands will have the force and stroke of a natural muscle and is capable of mimicking the mobility of human hands. This will make gripping not only easier but smarter and safer. By introducing the advancements in materials as well as technologies in sensing and actuation, it is expected that the performance and flexibility of robot grippers to be increased.

Grippers with mechanical simplicity and robustness (e.g., dual arm gripper) *versus* grippers with flexibility and adaptability (e.g., artificial hands) are two extreme objectives in robotic gripping. Simple mechanical grippers require simple control architecture and are widely used because of low price and simplicity of implementation. Besides, their applications become limited when it comes to grasping objects with force control, grasping objects with odd shapes, or specific grasping conditions. In such

applications, use of graspers with more adaptability like multi-fingered grippers becomes beneficial with the cost of control architecture complexity.

Recent advancements and applications indicate that soft grippers are one of the frontiers in the future in robot grippers for many applications. The emerging applications are mostly in industry and medical. Employing these advancements in industry will improve the performance significantly as reported; however the cost of changing the current technology and update them with the recent advancements is high. In medical applications especially in robotic surgery research is still going to provide surgeries with safe, robust and most importantly reliable mechanism.

5. Conclusions

End-effectors or grippers are very versatile components that have been used in applications such as automotive industries, manufacturing lines, vegetable pickers, MEMS grasping, surgical applications and prosthetic arms. Most of these grippers utilize an impactive gripping strategy. Since then, impactive grippers have branched into many segments including grippers that have added sensors, added cameras, piezoelectric grippers, soft grippers, MEMS, multi-fingered grippers, under actuated grippers, elastic and malleable grippers, and multi-arm grippers. Overall, grippers that have more sensor feedback tend to track and grasp objects more frequently. Piezoelectric and MEMS grippers actuate with high precision, however they do not have much grip strength. Malleable grippers can pick up many oddly shaped objects, however they may deteriorate after many cycles of actuation. Dual and multi-arm grippers are trending to replace humans in many tasks such as assembly lines and healthcare.

We discussed that how the performance of the grippers can be improved in future in different applications through robust controllers, instrumentations as well as structure design. We analyzed the implementation of human gripping capabilities on robotic grippers by employing vision sensing, incorporating visual feedback, artificial muscles, and smart and soft materials. We explored detailed advancements of performance and flexibility of robotic grippers through different designs and applications. The challenges in implementation different designs are discussed. We showed that soft, modular and adaptive grippers by utilizing different gripping components and materials can increase both performance and flexibility; however, as we argued the challenges in these improvements, introducing softness into the robotic gripper design requires set of design and control principles compared to hard grippers, and smart materials are not capable of providing strong gripping force.

This paper covered the earliest types of robotic grippers, the improvements made to them and recent advancements along with applications. We hope that by comparing and categorizing these robotic grippers under certain applications, new innovative methods can be discovered to solve real world problems present newer challenges that can be investigated to further push the boundaries in gripping mechanisms.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Siciliano, B.; Khatib, O. Springer Handbook of Robotics; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2008.
- 2. Monkman, G.J.; Hesse, S.; Steinmann, R.; Schunk, H. *Robot Grippers*; John Wiley & Sons: Weinheim, Germany, 2007.
- 3. Krüger, J.; Lien, T.K.; Verl, A. Cooperation of human and machines in assembly lines. *CIRP Ann. Manuf. Technol.* **2009**, *58*, 628–646. [CrossRef]
- 4. Fantoni, G.; Santochi, M.; Dini, G.; Tracht, K.; Scholz-Reiter, B.; Fleischer, J.; Lien, T.K.; Seliger, G.; Reinhart, G.; Franke, J.; *et al.* Grasping devices and methods in automated production processes. *CIRP Ann. Manuf. Technol.* **2014**, *63*, *679*–701. [CrossRef]
- 5. Staretu, I. *Gripping Systems*; Derc Publishing House: Tewksbury, MA, USA, 2011.

Robotics **2016**, 5, 11 15 of 20

Patel, Y.D.; George, P.M. Parallel Manipulators Applications—A Survey. Mod. Mech. Eng. 2012, 2, 57–64.
 [CrossRef]

- 7. Bertelsen, A.; Melo, J.; Sánchez, E.; Borro, D. A review of surgical robots for spinal interventions. *Int. J. Med. Robot. Comput. Assist. Surg.* **2013**, *9*, 407–422. [CrossRef] [PubMed]
- 8. Smith, B.; Karayiannidis, Y.; Nalpantidis, L.; Gratal, X.; Qi, P.; Dimarogonas, D.V.; Kragic, D. Dual arm manipulation—A survey. *Robot. Auton. Syst.* **2012**, *60*, 1340–1353. [CrossRef]
- 9. Hirzinger, G.; Brunner, B.; Landzettel, K.; Schott, J. Preparing a new generation of space robots—A survey of research at DLR. *Robot. Auton. Syst.* **1998**, 23, 99–106. [CrossRef]
- 10. Naoshi, K.; Ting, K.C. Robotics for Plant Production. Artif. Intell. Rev. 1998, 12, 227–243.
- 11. Savia, M.; Kiovo, H.N. Contact Micromanipulation—Survey of Strategies. *IEEE/ASME Trans. Mechatron.* **2009**, *14*, 504–514. [CrossRef]
- 12. Chiaverini, S.; Siciliano, B.; Villani, L. A survey of robot interaction control schemes with experimental comparison. *IEEE/ASME Trans. Mechatron.* **1999**, *4*, 273–285. [CrossRef]
- 13. Still In Motion. Kuka Robot/Inhouse. 2015. Available online: http://www.stillinmotion.de/portfolio/kuka-roboter/ (accessed on 4 June 2015).
- 14. OnOrbit. Canadarm2 and Dextre. 6 October 2014. Available online: http://spaceref.com/onorbit/canadarm2-and-dextre.html (accessed on 4 June 2015).
- 15. iRobot. iRobot 510 PackBot. 2015. Available online: http://www.irobot.com/For-Defense-and-Security/Robots/510-PackBot.aspx#Hazmat (accessed on 4 June 2015).
- 16. Sutter Health. Single-Site™ Instrumentation for the da Vinci® Si™ Surgical System. 2015. Available online: http://www.altabatessummit.org/clinical/robotic-surgery/ (accessed on 4 June 2015).
- 17. Chelpanov, I.B.; Kolpashnikov, S.N. Problems with the mechanics of industrial robot grippers. *Mech. Mach. Theory* **1983**, *18*, 295–299. [CrossRef]
- 18. Chen, F.Y. Force analysis and design considerations of grippers. Ind. Rob. Int. J. 1982, 9, 243–249. [CrossRef]
- 19. Devol, G.J.C. Programmed Article Transfer. U.S. Patent 2,988,237, 13 June 1961.
- 20. Totsuka, H. Manipulator. U.S. Patent 3,739,923, 17 February 1971.
- 21. Burton, C. Article Forming Machine and Extractor. U.S. Patent 3,765,474, 14 June 1971.
- 22. Victor, A.C.; Ronald, H.C. Tensor Arm Manipulator. U.S. Patent 3,497,083, 10 May 1968.
- 23. Ellwood, R.; Raatz, A.; Hesselbach, J. Vision and Force Sensing to Decrease Assembly Uncertainty. In *Precision Assembly Technologies and Systems*; Springer Berlin Heidelberg: Chamonix, France, 2010; pp. 123–130.
- 24. Schunk. 2-Finger Parallel Grippers. 2015. Available online: http://www.ca.schunk.com/schunk/schunk_websites/products/products_level_3/product_level3.html?product_level_3=289&product_level_2=250&product_level_1=244&country=CAN&lngCode=EN&lngCode2=EN# (accessed on 4 June 2015).
- 25. Active Robots. Robotiq 2-Finger Adaptive Gripper 200. 2015. Available online: http://www.active-robots.com/robotiq-2-finger-adaptive-gripper-200 (accessed on 4 June 2015).
- 26. Tracht, K.; Hogreve, S.; Bosse, S. Intrepretation of Multiaxial Gripper Force Sensors. In Proceedings of the 4th CIRP Conference on Assembly Technologies and Systems, Ann Arbor, MI, USA, 21–22 May 2012.
- 27. Hogreve, S.; Tracht, K. Design and implementation of multiaxial force sensing gripper fingers. *Prod. Eng.* **2014**, *8*, 765–772. [CrossRef]
- 28. Issa, M.; Petkovic, D.; Pavlovic, N.; Zentner, L. Sensor elements made of conductive silicone rubber for passively compliant gripper. *Int. J. Adv. Manuf. Technol.* **2013**, *69*, 1527–1536. [CrossRef]
- Bosse, S.; Hogreve, S.; Tracht, K. Design of a Mechanical Gripper with an Integrated Smart Sensor Network for Multi-Axial Force Sensing and Perception of Environment. In Proceedings of the Smart Systems Integration Conference 2012, Zürich, Switzerland, 21–22 March 2012.
- 30. Ulmen, J.; Cutkosky, M. A Robust, Low-Cost and Low-Noise Artificial Skin for Human-Friendly Robots. In Proceedings of the IEEE International Conference on Robotics and Automation, Anchorage, AK, USA, 3–7 May 2010.
- 31. Wettels, N.; Parnandi, A.R.; Moon, J.-H.; Loeb, G.E.; Sukhatme, G.S. Grip Control Using Biomimetic Tactile Sensing Systems. *IEEE/ASME Trans. Mechatron.* **2009**, *14*, 718–723. [CrossRef]
- 32. Drimus, A.; Kootstra, G.; Bilberg, A.; Kragic, D. Design of a flexible tactile sensor for classification of rigid and deformable objects. *Robot. Auton. Syst.* **2014**, *62*, 3–15. [CrossRef]

33. Hujic, D.; Croft, E.A.; Zak, G.; Fenton, R.G.; Mills, J.K.; Benhabib, B. The Robotic Interception of Moving Objects in Industrial Settings: Strategy Development and Experiment. *IEEE/ASME Trans. Mechatron.* **1998**, *3*, 225–239. [CrossRef]

- 34. Tsuji, T.; Kaneko, Y.; Abe, S. Whole-Body Force Sensation by Force Sensor With Shell-Shaped End-Effector. *IEEE Trans. Ind. Electron.* **2009**, *56*, 1375–1382. [CrossRef]
- 35. Arisumi, H.; Yokoi, K.; Komoriya, K. Casting Manipulation—Midair Control of a Gripper by Impulsive Force. *IEEE Trans. Robot.* **2008**, 24, 402–415. [CrossRef]
- 36. Zahraee, A.H.; Paik, J.K.; Szewczyk, J.; Morel, G. Toward the Development of a Hand-Held Surgical Robot for Laparoscopy. *IEEE/ASME Trans. Mechatron.* **2010**, *15*, 853–861. [CrossRef]
- 37. Dumlu, A.; Erenturk, K. Trajectory Tracking Control for a 3-DOF Parallel Manipulator Using Fractional-Order PIλDμ Control. *IEEE Trans. Ind. Electron.* **2014**, *61*, 3417–3426. [CrossRef]
- 38. Sato, M.; Toda, M. Robust Motion Control of an Oscillatory-Base Manipulator in a Global Coordinate System. *IEEE Trans. Ind. Electron.* **2015**, *62*, 1163–1174. [CrossRef]
- 39. Kampmann, P.; Kirchner, F. Towards a fine-manipulation system with tactile feedback for deep-sea environments. *Robot. Auton. Syst.* **2015**, *67*, 115–121. [CrossRef]
- 40. Yang, C.; Qu, Z.; Han, J. Decoupled-Space Control and Experimental Evaluation of Spatial Electrohydraulic Robotic Manipulators Using Singular Value Decomposition Algorithms. *IEEE Trans. Ind. Electron.* **2014**, *61*, 3427–3438. [CrossRef]
- 41. Elizarov, A.I.; Shein, N.G. The use of production robots in the production of plate heat exchangers. *Chem. Pet. Eng.* **1982**, *18*, 54–56. [CrossRef]
- 42. Mantriota, G. Theoretical model of the grasp with vacuum gripper. *Mech. Mach. Theory* **2007**, 42, 2–17. [CrossRef]
- 43. Callies, R.; Fronz, S. Recursive modeling and control of multi-link manipulators with vacuum grippers. *Math. Comput. Simul.* **2008**, *79*, 906–916. [CrossRef]
- 44. Monkman, G.J. Automated handling of packaging materials. Ind. Robot 1993, 20, 16.
- 45. Connolly, C. Robots at the heart of Schubert packaging machinery lead to great flexibility. *Ind. Robot* **2007**, *34*, 277–280.
- 46. Karbassi, J. Getting a grip with suction cups. *Mach. Des.* **2009**, *81*, 52–54.
- 47. Yu, S.; Gil, M. Manipulator handling device for assembly of large-size panels. *Assem. Autom.* **2012**, 32, 361–372. [CrossRef]
- 48. Angelillo, D.J.; Park, J. Efficient robotic packing speeds soft drinks manufacture. *Ind. Rob.* 2002, 29, 272–274.
- 49. McKeown, C.; Webb, P. A reactive reconfigurable tool for aerospace structures. *Assem. Autom.* **2011**, *31*, 334–343. [CrossRef]
- 50. Lovell, A. Optimized vacuum system design improves productivity. Control Eng. 2006, 53, 26–28.
- 51. Kelley, R.B.; Birk, J.R.; Martins, H.A.S.; Tella, R. A Robot System Which Acquires Cylindrical Workpieces from Bins. *IEEE Trans. Syst.* **1982**, 12, 204–213. [CrossRef]
- 52. Sujan, V.A.; Dubowsky, S. Robotic Manipulation of Highly Irregular Shaped Objects: Application to a Robot Crucible Packing System for Semiconductor Manufacture. *J. Manuf. Process.* **2002**, *4*, 1–15. [CrossRef]
- 53. Wang, Y.; Zhang, G.-L.; Lang, H.; Zuo, B.; de Silva, C.W. A modified image-based visual servo controller with hybrid camera configuration for robust robotic grasping. *Robot. Auton. Syst.* **2014**, *62*, 1398–1407. [CrossRef]
- 54. Al-Hujazi, E.; Sood, A. Range Image Segmentation with Applications to Robot Bin-Picking Using Vacuum Gripper. *IEEE Trans. Syst. Man Cybern.* **1990**, *20*, 1313–1325. [CrossRef]
- 55. Keshmiri, M.; Xie, W.-F.; Mohebbi, A. Augmented Image-Based Visual Servoing of a Manipulator Using Acceleration Command. *IEEE Trans. Ind. Electron.* **2014**, *61*, 5444–5452. [CrossRef]
- 56. Du, G.; Zhang, P.; Li, D. Human-Manipulator Interface Based on Multisensory Process via Kalman Filters. *IEEE Trans. Ind. Electron.* **2014**, *61*, 5411–5418.
- 57. Serón, J.; Martínez, J.L.; Mandow, A.; Reina, A.J.; Morales, J.; García-Cerezo, A.J. Automation of the Arm-Aided Climbing Maneuver for Tracked Mobile Manipulators. *IEEE Trans. Ind. Electron.* **2014**, *61*, 3638–3647. [CrossRef]
- 58. Eizicovits, D.; Berman, S. Efficient sensory-grounded grasp pose quality mapping for gripper. *Robot. Auton. Syst.* **2014**, *62*, 1208–1219. [CrossRef]
- 59. Dearden, R.; Burbridge, C. Manipulation planning using learned symbolic state abstractions. *Robot. Auton. Syst.* **2014**, *62*, 355–365. [CrossRef]

Robotics **2016**, 5, 11 17 of 20

60. Cho, S.I.; Chang, S.J.; Kim, Y.Y.; An, K.J. Development of a Three-degrees-of-freedom Robot for harvesting Lettuce using Machine Vision and Fuzzy logic Control. *Biosyst. Eng.* **2002**, *82*, 143–149. [CrossRef]

- 61. Pettersson, A.; Ohlsson, T.; Davis, S.; Gray, J.O.; Dodd, T.J. A hygienically designed force gripper for flexible handling of variable and easily damaged natural food products. *Innov. Food Sci. Emerg. Technol.* **2011**, 12, 344–351. [CrossRef]
- 62. Sun, Q.; Zou, X.; Zou, H.; Chen, Y.; Cai, W. Intelligent Design and Kinematics Analysis of Picking Robot Manipulator. In Proceedings of the International Conference on Measuring Technology and Mechatronics Automation, Changsha, China, 1 January 2010.
- 63. Hayashi, S.; Yamamoto, S.; Shigematsu, K.; Kobayashi, K.; Kohno, Y.; Kamata, J.; Kurita, M. Performance of movable-type harvesting robot for strawberries. In Proceedings of the International Symposium on High Technology for Greenhouse Systems: GreenSys2009, Québec City, QC, Canada, 14–19 June 2009.
- 64. Mehta, S.S.; Burks, T.F. Vision-based control of robotic manipulator for citrus harvesting. *Comput. Electron. Agric.* **2014**, *102*, 146–158. [CrossRef]
- 65. PnuematicTiPS. Robots Learn to Pick the Harvest. 6 March 2015. Available online: http://www.pneumatictips.com/3965/2015/03/featured/robots-learn-to-pick-the-harvest/ (accessed on 4 June 2015).
- 66. Font, D.; Pallejà, T.; Tresanchez, M.; Runcan, D.; Moreno, J.; Martínez, D.; Teixidó, M.; Palacín, J. A Proposal for Automatic Fruit Harvesting by Combining a Low Cost Stereovision Camera and a Robotic Arm. *Sensors* **2014**, *14*, 11557–11579. [CrossRef] [PubMed]
- 67. Davis, S.; Gray, J.O.; Caldwell, D.G. An end effector based on the Bernoulli principle for handling sliced fruit and vegetables. *Robot. Comput.-Integr. Manuf.* **2008**, 24, 249–257. [CrossRef]
- 68. Blanes, C.; Cortes, V.; Ortiz, C.; Mellado, M.; Talens, P. Non-Destructive Assessment of Mango Firmness and Ripeness Using a Robotic Gripper. *Food Bioprocess Technol.* **2015**, *8*, 1914–1924. [CrossRef]
- 69. Monta, M.; Kondo, N.; Ting, K.C. End-Effectors for Tomato Harvesting Robot. *Artif. Intell. Rev.* **1998**, 12, 11–25. [CrossRef]
- 70. Chiu, Y.-C.; Yang, P.-Y.; Chen, S. Development of the End-Effector of a Picking Robot for Greenhouse-Grown Tomatoes. *Appl. Eng. Agric.* **2013**, *29*, 1001–1009.
- 71. Tanigaki, K.; Fujiura, T.; Akase, A.; Imagawa, J. Cherry-harvesting robot. *Comput. Electron. Agric.* **2008**, *63*, 65–72. [CrossRef]
- 72. Frank, B.; Stachniss, C.; Schmedding, R.; Teschner, M.; Burgard, W. Learning object deformation models for robot motion planning. *Robot. Auton. Syst.* **2014**, *62*, 1153–1174. [CrossRef]
- 73. Rateni; Cianchetti, M.; Ciuti, G.; Menciassi, A.; Laschi, C. Design and Development of a soft robotic gripper for manipulation in minimally invasive surgery: A proof of concept. *Meccanica* **2015**, *50*, 2855–2863. [CrossRef]
- 74. Lambercy, O.; Metzger, J.C.; Santello, M.; Gassert, R. A method to study precision grip control in viscoelastic force fields using a robotic gripper. *IEEE Trans. Biomed. Eng.* **2015**, *62*, 39–48. [CrossRef] [PubMed]
- 75. Gultepe, E.; Randhawa, J.S.; Kadam, S.; Yamanaka, S.; Selaru, F.M.; Shin, E.J.; Kalloo, A.N.; Gracias, D.H. Biopsy with Thermally-Responsive Untethered Microtools. *Adv. Mater.* **2013**, 25, 514–519. [CrossRef] [PubMed]
- 76. Vonck, D.; Jakimowicz, J.J.; Lopuhaä, H.P.; Goossens, R.H. Grasping soft tissue by means of vacuum technique. *Med. Eng. Phys.* **2012**, *34*, 1088–1094. [CrossRef]
- 77. Tortora, G.; Ranzani, T.; Falco, I.D.; Dario, P.; Menciassi, A. A miniature robot for retraction tasks under vision assistance in minimally invasive surgery. *Robotics* **2014**, *3*, 70–82. [CrossRef]
- 78. Kim, U.; Lee, D.H.; Yoon, W.J.; Hannaford, B.; Choi, H.R. Force Sensor Integrated Surgical Forceps for Minimally Invasive Robotic Surgery. *IEEE Trans. Robot.* **2015**, *31*, 1214–1224. [CrossRef]
- 79. Jin, H.L.; Delgado-Martinez, I.; Chen, H.Y. Customizable Soft Pneumatic Chamber-Gripper Devices for Delicate Surgical Manipulation. *J. Med. Devices* **2014**, *8*, 044504.
- 80. Ullrich, F.; Dheman, K.S.; Schuerle, S.; Nelson, B.J. Magnetically actuated and guided milli-gripper for medical applications. In Proceedings of the IEEE International Conference on Robotics and Automation (ICRA), Seattle, WA, USA, 26–30 May 2015.
- 81. Dechev, N.; Cleghorn, W.L.; Mills, J.K. Microassembly of 3D microstructures using a compliant, passive microgripper. *J. Microelectromech. Syst.* **2004**, *13*, 176–189. [CrossRef]

82. Fatikow, S.; Eichhorn, V.; Jasper, D.; Weigel-Jech, M.; Niewiera, F.; Krohs, F. Automated Nanorobotic Handling of Bio- and Nano-Materials. In Proceedings of the 6th Annual IEEE Conference on Automation Science and Engineering, Toronto, ON, Canada, 21–24 August 2010.

- 83. Chen, L.; Liu, B.; Chen, T.; Shao, B. Design of Hybrid-type MEMS Microgripper. In Proceedings of the IEEE International Conference on Mechatronics and Automation, Changchun, China, 9–12 August 2009.
- 84. Wang, X.; Vincent, L.; Yu, M.; Huang, Y.; Liu, C. Architecture of a Three-Probe MEMS Nanomanipulator with Nanoscale End-Effectors. In Proceedings of the International Conference on Advanced Intelligent Mechatronics, Kobe, Japan, 20–24 July 2003.
- 85. Millet, O.; Bernardoni, P.; Régnier, S.; Bidaud, P.; Tsitsiris, E.; Collard, D.; Buchaillot, L. Electrostatic actuated micro gripper using an amplification mechanism. *Sens. Actuators A Phys.* **2004**, *114*, 371–378. [CrossRef]
- 86. Chen, B.K.; Zhang, Y.; Perovic, D.D.; Sun, Y. MEMS microgrippers with thin gripping tips. *J. Micromech. Microeng.* **2011**, 21, 1–5. [CrossRef]
- 87. Demaghsi, H.; Mirzajani, H.; Ghavifekr, H.B. A novel electrostatic based microgripper (cellgripper) integrated with contact sensor and equipped with vibrating system to release particles actively. *Microsyst. Technol.* **2014**, *20*, 2191–2202. [CrossRef]
- 88. Mackay, R.E.; Le, H.R.; Clark, S.; Williams, J.A. Polymer micro-grippers with an integrated force sensor for biological manipulation. *J. Micromech. Microeng.* **2013**, 23, 1–7. [CrossRef]
- 89. Avci, E.; Ohara, K.; Nguyen, C.-N.; Theeravithayangkura, C.; Kojima, M.; Tanikawa, T.; Mae, Y.; Arai, T. High-Speed Automated Manipulation of Microobjects Using a Two-Fingered Microhand. *IEEE Trans. Ind. Electron.* **2015**, *62*, 1070–1079. [CrossRef]
- 90. Fontana, G.; Ruggeri, S.; Fassi, I. A mini work-cell for handling and assembling microcomponents. *Assem. Autom.* **2014**, *34*, 27–33. [CrossRef]
- 91. Hazra, S.S.; Beuth, J.L.; Myers, G.A.; DelRio, F.W.; de Boer, M.P. Design and test of reliable high strength ingressive polycrystalline silicon microgripper arrays. *J. Micromech. Microeng.* **2015**, 25, 1–12. [CrossRef]
- 92. Myers, G.A.; Hazra, S.S.; de Boer, M.P.; Michaels, C.A.; Stranick, S.J.; Koseski, R.P.; Cook, R.F.; DelRio, F.W. Stress mapping of micromachined polycrystalline silicon devices via confocal Raman microscopy. *Appl. Phys. Lett.* **2014**, *104*, 191908. [CrossRef]
- 93. Biganzoli, F.; Fantoni, G. A self-centering electrostatic microgripper. *J. Manuf. Syst.* **2008**, 27, 136–144. [CrossRef]
- 94. Liu, Y.; Xu, M.; Cao, Y. Research, design and experiment of end effector for wafer transfer robot. *Ind. Robot* **2012**, *39*, 79–91. [CrossRef]
- 95. Gauthier, M.; Réginer, S.; Lopez-Walle, B.; Gibeau, E.; Rougeot, P.; Hériban, D.; Chaillet, N. Micro-assembly and modeling of the liquid microworld: The PRONOMIA project. In Proceedings of the IEEE Workshop on Robotic Assembly of 3D MEMS IROS 2007, San Diego, CA, USA, 29 October–2 November 2007.
- 96. Walle, B.L.; Gauthier, M.; Chaillet, N. Submerged Freeze Gripper to Manipulate Micro-objects. In Proceedings of the International Conference on Intelligent Robots and Systems, Beijing, China, 9–15 October 2006.
- 97. Walle, B.L.; Gauthier, M.; Chaillet, N. Principle of a Submerged Freeze Gripper for Microassembly. *IEEE Trans. Robot.* **2008**, 24, 897–902. [CrossRef]
- 98. Walle, B.L.; Gauthier, M.; Chaillet, N. Dynamic modelling for a submerged freeze microgripper using thermal networks. *J. Micromech. Microeng.* **2010**, *20*, 1–10.
- 99. Sarhadi, M. A novel manpulator for automated handling of flexible materials. In Proceedings of the 2nd Duisburger Kolloquium Automation and Robotik, Duisberg, Geramny, 15–17 July 1987.
- 100. Schulz, G. Grippers for flexible textiles. In Proceedings of the 5th International Conference on Advanced Robotics, Pisa, Italy, 20–22 June 1991.
- 101. Sarhadi, M. Robotic handling and lay-up advanced composite materials: An overview. In *Sensory Robotics for the Handling of Limp Materials*; Springer-Verlag: New York, NY, USA, 1990; pp. 33–50.
- 102. Monkman, G.J. Robot Grippers for Use With Fibrous Materials. *Int. J. Robot. Res.* **1995**, *14*, 144–151. [CrossRef]
- 103. Jarvis, S.D.H.; Wilcox, K.; Chen, X.Q.; McCarthy, R.; Sarhadi, M. Design of a handling device for composite ply lay-up automation. In Proceedings of the IEEE 5th International Conference on Advanced Robotics, Pisa, Italy, 20–22 June 1991.

104. Fleischer, J.; Ochs, A.; Dosch, S. The Future of Lightweight Manufacturing—Production-related Challenges When Hybridizing Metals and Continuous Fiber-reinforced Plastics. In Proceedings of the International Conference on New Developments in Sheet Metal Forming, Stuttgart, Germany, 22–23 May 2012.

- 105. Kolluru, R.; Valavanis, K.P.; Steward, A.; Sonnier, M.J. A Flat-Surface Robotic Gripper for Handling Limp Material. *IEEE Robot. Autom. Mag.* **1995**, *2*, 19–26. [CrossRef]
- 106. Kolluru, R.; Valavanis, K.P.; Hebert, T.M. Modeling, Analysis, and Performance Evaluation of A Robotic Gripper System for Limp Material Handling. *IEEE Trans. Syst. Man Cybern. B* **1998**, 28, 480–486. [CrossRef] [PubMed]
- 107. Reinhart, G.; Straßer, G.; Ehinger, C. Highly Flexible Automated Manufacturing of Composite Structures Consisting of Limp Carbon Fibre Textiles. *SAE Int. J. Aerosp.* **2010**, *2*, 181–187. [CrossRef]
- 108. Reinhart, G.; Ehinger, C. Novel Robot-Based End-Effector Design for an Automated Preforming of Limb Carbon Fiber Textiles. In *Future Trends in Production Engineering*; Springer: Berlin/Heidelberg, Germany, 2013; pp. 131–142.
- 109. Dini, G.; Failli, F.; Sebastiani, F. Development of Automated Systems for Manipulation and Quality Control of Natural Leather Plies. 10 February 2005. Available online: http://www2.ing.unipi.it/leather_project/vacuum_cup.htm (accessed on 29 March 2015).
- 110. Monkman, G.J.; Shimmin, C. Robot grippers using Permatack adhesives. *Assem. Autom.* **1991**, *11*, 17–19. [CrossRef]
- 111. Ruth, D.E.; Mulgaonkar, P. Robotic lay-up of prepeg composite plies. In Proceedings of the IEEE International Conference on Robotics and Automation, Cincinnati, OH, USA, 13–18 May 1990.
- 112. Raatz, A.; Rathmann, S.; Hesselbach, J. Process development for the assembly of microsystems with hot melt adhesives. *CIRP Ann. Manuf. Technol.* **2012**, *61*, 5–8. [CrossRef]
- 113. Monkman, G. Precise piezoelectric prehension. Ind. Robot 2000, 27, 189-194. [CrossRef]
- 114. Tzou, H.S. Development of a light-weight robot end-effector using polymeric piezoelectric bimorph. In Proceedings of the IEEE International Conference on Robotics and Automation, Scottsdale, AZ, USA, 14–19 May 1989.
- 115. Kurita, Y.; Sugihara, F.; Ueda, J.; Ogasawara, T. Piezoelectric Tweezer-Type End Effector With Force- and Displacement-Sensing Capability. *IEEE/ASME Trans. Mechatron.* **2012**, *17*, 1039–1048. [CrossRef]
- 116. McPherson, T.; Ueda, J. A Force and Displacement Self-Sensing Piezoelectric MRI-Compatible Tweezer End Effector with an On-Site Calibration Procedure. *IEEE/ASME Trans. Mechatron.* **2014**, *19*, 755–764. [CrossRef]
- 117. Jain, R.K.; Majumder, S.; Ghosh, B. Design and analysis of piezoelectric actuator for micro gripper. *Int. J. Mech. Mater. Des.* **2014**, *11*, 253–276. [CrossRef]
- 118. Jain, R.K.; Majumder, S.; Ghosh, B.; Saha, S. Design and manufacturing of mobile micro manipulation system with a compliant piezoelectric actuator based micro gripper. *J. Manuf. Syst.* **2015**, *35*, 76–91. [CrossRef]
- 119. Li, J.; Yang, L. Adaptive PI-Based Sliding Mode Control for Nanopositioning of Piezoelectric Actuators. *Math. Probl. Eng.* **2014**, 2014, 257864. [CrossRef]
- 120. El-Sayed, A.M.; Abo-Ismail, A.; El-Melegy, M.T.; Hamzaid, N.A.; Osman, N.A.A. Development of a Micro-Gripper Using Piezoelectric Bimorphs. *Sensors* **2013**, *13*, 5826–5840. [CrossRef] [PubMed]
- 121. Crossley, F.R.E.; Umholtz, F.G. Design for a Three-fingered Hand. *Mech. Mach. Theory* **1977**, *12*, 85–93. [CrossRef]
- 122. Konno, A.; Tada, M.; Nagashima, K.; Inaba, M.; Inoue, H. Development of a 3-Fingered Hand and Grasping Unknown Objects by Groping. In Proceedings of the IEEE International Symposium on Assembly and Task Planning, Marina del Rey, CA, USA, 7–9 August 1997.
- 123. Robotiq. Adaptive Robot Gripper 3-Finger. 2015. Available online: http://robotiq.com/products/industrial-robot-hand/ (accessed on 4 June 2015).
- 124. Prensilia s.r.l. Robotic Hands (Self-Contained). 2014. Available online: http://www.prensilia.com/index. php?q=en/node/40 (accessed on 4 June 2015).
- 125. Shadow Robot Company. Shadow Dexterous HandTM—Now Available for Purchase! 2015. Available online: http://www.shadowrobot.com/products/dexterous-hand/ (accessed on 4 June 2015).
- 126. Wakimoto, S.; Ogura, K.; Suzumori, K.; Nishioka, Y. Miniature Soft Hand with Curling Rubber Pneumatic Actuators. In Proceedings of the IEEE International Conference on Robotics and Automation, Kobe, Japan, 12–17 May 2009.

Robotics **2016**, 5, 11 20 of 20

127. Cianchetti, M.; Laschi, C. An Under-Actuated and Adaptable Soft Robotic Gripper. In Proceedings of the 4th International Conference on Living Machines, Barcelona, Spain, 28–31 July 2015.

- 128. Hu, L.; Wang, Y.; Zhang, J.; Cui, Y.; Ma, L.; Jiang, J.; Fang, L.; Zhang, B. A massage robot based on Chinese massage therapy. *Ind. Robot* 2013, 40, 158–172. [CrossRef]
- 129. Hoffmann, H.; Chen, Z.; Earl, D.; Mitchell, D.; Salemi, B.; Sinapov, J. Adaptive robotic tool use under variable grasps. *Robot. Auton. Syst.* **2014**, *62*, 833–846. [CrossRef]
- 130. Odhner, L.U.; Dollar, A.M. Dexterous Manipulation with Underactuated Elastic Hands. In Proceedings of the IEEE International Conference on Robotics and Automation, Shanghai, China, 9–11 May 2011.
- 131. Kragten, G.A.; Baril, M.; Gosselin, C.; Herder, J.L. Stable Precision Grasps by Underactuated Grippers. *IEEE Trans. Robot.* **2011**, 27, 1056–1066. [CrossRef]
- 132. Fu, Z.; Zhou, H.; Liu, Z.; Fei, J.; Yan, W.; Zhao, Y. Design of a Robot End-Effector Grabbing Mechanism Based on a Bionic Snake Mouth. *Int. J. Adv. Robot. Syst.* **2012**, *10*. [CrossRef]
- 133. Hatakeyama, T.; Mochiyama, H. Shooting Manipulation Inspired by Chameleon. *IEEE/ASME Trans. Mechatron.* **2013**, *18*, 527–535. [CrossRef]
- 134. Ciocarlie, M.; Hicks, F.M.; Holmberg, R.; Hawke, J.; Schlicht, M.; Gee, J.; Stanford, S.; Bahadur, R. The Velo gripper: A versatile single-actuator design for enveloping, parallel and fingertip grasps. *Int. J. Robot. Res.* **2014**, *33*, 753–767. [CrossRef]
- 135. Shapiro, Y.; Wolf, A.; Gabor, K. Bi-bellows: Pneumatic bending actuator. *Sens. Actuators A Phys.* **2011**, *167*, 484–494. [CrossRef]
- 136. Barber, J.; Volz, R.A.; Desai, R.; Rubinfeld, R.; Schipper, B.D.; Wolter, J. Automatic Evaluation of Two-Fingered Grips. *IEEE J. Robot. Autom.* **1987**, *3*, 356–361. [CrossRef]
- 137. Doulgeri, Z.; Fasoulas, J. Grasping control of rolling maniplations with deformable fingertips. *IEEE/ASME Trans. Mechatron.* **2003**, *8*, 283–286. [CrossRef]
- 138. Kenaley, G.L.; Cutkosky, M.R. Electrorheological fluid-based robotic fingers with tactile sensing. In Proceedings of the IEEE Conference on Robotics and Automation, Scottsdale, AZ, USA, 14–19 May 1989.
- 139. Brown, E.; Rodenberg, N.; Amend, J.; Mozeika, A.; Steltz, E.; Zakin, M.R.; Lipson, H.; Jaeger, H.M. Universal robotic gripper based on the jamming of granular material. *Proc. Natl. Acad. Sci. USA* **2010**, *107*, 18809–18814. [CrossRef]
- 140. Amend, J.R.; Brown, E.; Rodenberg, N.; Jaeger, H.M.; Lipson, H. A Positive Pressure Universal Gripper Based on the Jamming of Granular Material. *IEEE Trans. Robot.* **2012**, *28*, 341–350. [CrossRef]
- 141. Empire Robotics. Versaball Advantage. 2015. Available online: http://empirerobotics.com/products.html (accessed on 4 June 2015).
- 142. Pettersson, A.; Davis, S.; Gray, J.O.; Dodd, T.J.; Ohlsson, T. Design of a magnetorheological robot gripper for handling of delicate food products with varying shapes. *J. Food Eng.* **2010**, *98*, 332–338. [CrossRef]
- 143. Madhani, J.; Salisbury, J.K. Articulated Surgical Instrument for Performing Minimally Invasive Surgery with Enhanced Dexterity and Sensitivity. U.S. Patent 5,792,135, 11 August 1998.



© 2016 by the authors; licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC-BY) license (http://creativecommons.org/licenses/by/4.0/).